\begin{table}%t2
%\centering
\par
\caption{\label{t:doubles}Repeated entries in \hevelius.}
\begin{tabular}{rrrrcl}
\hline\hline\noalign{\smallskip}
\phantom{H}H & \phantom{H}H & HIP\phantom{HP} & $d_{1,2}$(\arcmin) &
$\Delta M$ & Figure: location\\
\hline
  16 & 1202 &   4463 &  1.0 & 0 & \ref{f:andromeda}: $-$4.4,$-$16.7 \\
 321 & 1460 &  63125 &  0.0 & 0 & \ref{f:canesven}: $-$4.1,$-$2.0 \\
 322 & 1488 &  61317 &  0.0 & 0 & \ref{f:canesven}: $-9.2$,$-$1.0 \\
 349 &  689 &  59504 &  0.2 & 0 & \ref{f:camelopardalis}: 11.4,17.3 \\
 692 & 1536 &  73199 &  9.1 & 1 & \ref{f:ursaminor}: 10.4,3.4 \\
 887 &  919 &  52457 &  0.4 & 1 & \ref{f:leo}: $-$1.6,6.7 \\
 889 &  916 &  53417 &  0.0 & 0 & \ref{f:leo}: 0.3,9.2 \\
 890 &  918 &  54951 &  0.0 & 0 & \ref{f:leo}: 5.0,9.5 \\
1091 & 1557 & 106140 &  0.0 & 0 & \ref{f:pegasus}: $-$14.5,11.4 \\
1145 & 1151 &  17313 & 15.7 & 1 & \ref{f:perseus}: 0.7,$-$11.0 \\
1150 & 1164 &  17886 &  0.0 & 0 & \ref{f:perseus}: 1.9,$-$12.1 \\
1154 & 1163 &  14382 &  0.0 & 0 & \ref{f:perseus}: 0.9,12.5 \\
1299 & 1326 &  80179 &  0.3 & 1 & \ref{f:serpens}: $-$12.7,5.7 \\
\hline
\end{tabular}
\end{table}